Author's response to reviews

Title: Intracranial hypotension secondary to spinal arachnoid cyst rupture presenting with acute severe headache: A case report

Authors:

Wendy Jones (wendy_d_jones@yahoo.co.uk)
Makrand Kulkarni (makarand_kul@yahoo.com)
K Ravishankar (dr.k.ravishankar@gmail.com)
Rudi Borgstein (rudi.borgstein@nmh.nhs.uk)
Peter Dupont (peter.dupont@nmh.nhs.uk)

Version: 3 Date: 1 November 2010

Author's response to reviews:

Dear Editorial Production Team,

Many thanks for your swift editing of our manuscript. As regards the changes you have suggested:

1. I have amended the authors so Wendy Jones no longer appears twice
2. I have checked all the authors email addresses
3. I have removed the header from the manuscript
4. I have changed lady to female
5. I have changed the acknowledgements section to include all authors reviewed the final manuscript

In addition I changed the designation of two of the authors I also changed the colour of some of the text from red to black.

I did not format the two figures and would be extremely grateful if the editorial team were able to format them. Please do not hesitate to contact me if you have any questions or would like to make any further changes. I am very excited about the publication of our case report.

Kind regards

Yours faithfully

Dr Wendy Jones